Author's response to reviews

Title: Using height-for-age difference (HAD) instead of height-for-age z-scores (HAZ) for the meaningful measurement of population-level catch-up in linear growth in children less than 5 years of age

Authors:

Jef L Leroy (j.leroy@cgiar.org)
Marie Ruel (m.ruel@cgiar.org)
Jean-Pierre Habicht (jh48@cgiar.org)
Edward A Frongillo (EFRONGILLO@sc.edu)

Version: 2 Date: 12 November 2014

Author's response to reviews:

New submission